% Created 2014-07-02 Wed 18:50
\documentclass{article}
\usepackage[utf8]{inputenc}
\usepackage[T1]{fontenc}
\usepackage{fixltx2e}
\usepackage{graphicx}
\usepackage{longtable}
\usepackage{float}
\usepackage{wrapfig}
\usepackage{rotating}
\usepackage[normalem]{ulem}
\usepackage{amsmath}
\usepackage{textcomp}
\usepackage{marvosym}
\usepackage{wasysym}
\usepackage{amssymb}
\usepackage{hyperref}
\tolerance=1000
\usepackage{color}
\usepackage{listings}
\author{Rainer M. Krug}
\date{\today}
\title{reftest}
\hypersetup{
 pdfkeywords={},
  pdfsubject={},
  pdfcreator={Emacs 24.3.1 (Org mode beta_8.3)}}
\begin{document}

\maketitle
\tableofcontents



\section{Introduction}
\label{sec-1}

\subsection{Here comes the test}
\label{sec-1-1}

Some test refs (\cite{Vallet2011,Paquette2011,Lebourgeois2013,Forrester2013}). And one more \cite{Vila2013}, plus one more  \cite{Charru2011}. 

\section{References}
\label{sec-2}
\bibliography{~/Documents/Projects/IFN/ProductivityMixedStands/prodMixStands}
% Emacs 24.3.1 (Org mode beta_8.3)
\end{document}